Author’s response to reviews

Title: The influence of behavioural and socioeconomic factors on the community injury rates of adolescents assessed by the South Korean Emergency Medical Services: An ecological approach

Authors:
Ki Ok Ahn (arendt75@gmail.com)
Jungeun Kim (ckck1121@hanmail.net)
Sang Do Shin (sdshin@snu.ac.kr)
Hyesook Park (hpark@ewha.ac.kr)
Federico Vaca (federico.vaca@yale.edu)
Ju Ok Park (juok.park@gmail.com)

Version: 1 Date: 10 Jun 2019

Author’s response to reviews:

Dear All reviewer:

Thank you very much for reviewing our paper. On behalf of all authors, I also greatly appreciate the reviewers for their complimentary comments and suggestions. The suggestions offered by the reviewers have been very helpful, and we also appreciate your insightful comments on revising the paper.

We have carried out the analysis that the reviewers suggested and revised the manuscript accordingly. Please find attached a point-by-point response to reviewer’s concerns. The changes are marked in yellow highlight in the paper, and the revised manuscript is uploaded.

We hope you are satisfied with our answer but are happy to consider further revisions, and we thank you for your continued interest in our research. We also hope that the manuscript is now acceptable for publication.

Sincerely,

Ju Ok, Park